UC Davis UC Davis Previously Published Works

Title

Changes in Natriuretic Peptide Levels and Subsequent Kidney Function Decline in SPRINT

Permalink https://escholarship.org/uc/item/04g361fw

Journal American Journal of Kidney Diseases, 83(5)

ISSN 0272-6386

Authors

Ascher, Simon B Berry, Jarett D Katz, Ronit <u>et al.</u>

Publication Date

2024-05-01

DOI

10.1053/j.ajkd.2023.09.018

Peer reviewed



HHS Public Access

Author manuscript *Am J Kidney Dis*. Author manuscript; available in PMC 2025 January 09.

Published in final edited form as:

Am J Kidney Dis. 2024 May; 83(5): 615–623.e1. doi:10.1053/j.ajkd.2023.09.018.

Changes in Natriuretic Peptide Levels and Subsequent Kidney Function Decline in SPRINT

Simon B. Ascher, MD, MPH,

Division of Hospital Medicine, University of California, Davis, Sacramento

Kidney Health Research Collaborative, Department of Medicine, San Francisco Veterans Affairs Health Care System and University of California San Francisco, San Francisco

Jarett D. Berry, MD, MS,

Divison of Cardiology, Department of Internal Medicine, University of Texas Southwestern Medical Center, Dallas, Texas

Ronit Katz, DPhil,

Department of Obstetrics and Gynecology, University of Washington, Seattle, Washington

James A. de Lemos, MD,

Divison of Cardiology, Department of Internal Medicine, University of Texas Southwestern Medical Center, Dallas, Texas

Nisha Bansal, MD, MAS,

Kidney Research Institute, Division of Nephrology, Department of Medicine, University of Washington, Seattle, Washington

Pranav S. Garimella, MBBS, MPH,

Financial Disclosure: Dr Shlipak has received consulting income from Cricket Health Inc; serves on advisory panels for Boehringer Ingelheim, AztraZeneca, and Bayer; and has received research support from Bayer. Dr lx holds an investigator initiated research grant from Baxter International Inc, serves as a member of a data safety monitoring board for Sanifit Therapeutics, is a member of the scientific advisory board for Alpha Young, and has served on advisory boards for AstraZeneca and Ardelyx. Dr Berry reports grant support from Roche Diagnostics and Abbott Diagnostics and consulting fees from Roche Diagnostics, AstraZeneca, and the Cooper Institute. Dr de Lemos reports grant support from Roche Diagnostics, Ortho Clinical Diagnostics, Beckman Coulter, Quidel Cardiovascular Inc, and Siemens Health Care Diagnostics; and has been named a coowner on a patent application awarded to the University of Maryland (US Patent Application Number 15/309,754) entitled "Methods for Assessing Differential Risk for Developing Heart Failure." The remaining authors declare that they have no relevant financial interests.

Disclaimer: The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health, the US Department of Veterans Affairs, or the United States government.

Peer Review: Received April 28, 2023. Evaluated by 2 external peer reviewers, with direct editorial input from a Statistics/Methods Editor, an Associate Editor, and a Deputy Editor who served as Acting Editor-in-Chief. Accepted in revised form September 17, 2023. The involvement of an Acting Editor-in-Chief was to comply with *AJKD's* procedures for potential conflicts of interest for editors, described in the Information for Authors & Journal Policies.

This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Address for Correspondence: Simon B. Ascher, MD, MPH, Division of Hospital Medicine, Department of Medicine, University of California, Davis, 2315 Stockton Blvd, Suite 2P101, Sacramento, CA 95817. sbascher@ucdavis.edu. *JHI and MGS contributed equally to this work.

Authors' Contributions: Research idea and study design: SBA, JDB, RK, JHI, MGS; data acquisition: JDB, JHI, MGS; data analysis: RK; data interpretation: SBA, JDB, RK, JAdL, NB, PSG, SIH, NW, VKJ, AAK, JHI, MGS; supervision or mentorship: JHI, MGS. Each author contributed important intellectual content during manuscript drafting or revision and agrees to be personally accountable for the individual's own contributions and to ensure that questions pertaining to the accuracy or integrity of any portion of the work, even one in which the author was not directly involved, are appropriately investigated and resolved, including with documentation in the literature if appropriate.

Division of Nephrology-Hypertension, University of California, San Diego, La Jolla

Stein I. Hallan, MD, PhD,

Department of Nephrology, St Olav University Hospital, Trondheim, Norway

Department of Clinical and Molecular Medicine, Faculty of Medicine, Norwegian University of Science and Technology, Trondheim, Norway

Nicholas Wettersten, MD,

Division of Cardiology, Department of Medicine, University of California, San Diego, La Jolla

Cardiology Section, Veterans Affairs San Diego Healthcare System, San Diego, California

Vasantha K. Jotwani, MD,

Kidney Health Research Collaborative, Department of Medicine, San Francisco Veterans Affairs Health Care System and University of California San Francisco, San Francisco

Anthony A. Killeen, MB, BCh, MSc, PhD,

Department of Laboratory Medicine and Pathology, University of Minnesota, Minneapolis, Minnesota

Joachim H. Ix, MD, MAS^{*},

Nephrology Section, Veterans Affairs San Diego Healthcare System, San Diego, California

Division of Nephrology-Hypertension, University of California, San Diego, La Jolla

Michael G. Shlipak, MD, MPH^{*}

Kidney Health Research Collaborative, Department of Medicine, San Francisco Veterans Affairs Health Care System and University of California San Francisco, San Francisco

Abstract

Rationale & Objective: Novel approaches to the assessment of kidney disease risk during hypertension treatment are needed because of the uncertainty of how intensive blood pressure (BP) lowering impacts kidney outcomes. We determined whether longitudinal N-terminal pro–B-type natriuretic peptide (NT–proBNP) measurements during hypertension treatment are associated with kidney function decline.

Study Design: Prospective observational study.

Setting & Participants: 8,005 SPRINT (Systolic Blood Pressure Intervention Trial) participants with NT–proBNP measurements at baseline and 1 year.

Exposure: 1-year change in NT–proBNP categorized as a 25% decrease, 25% increase, or <25% change (stable).

Outcome: Annualized change in estimated glomerular filtration rate (eGFR) and 30% decrease in eGFR.

Analytical Approach: Linear mixed-effect and logistic regression models were used to evaluate the association of changes in NT–proBNP with subsequent annualized change in eGFR and 30% decrease in eGFR, respectively. Analyses were stratified by baseline chronic kidney disease (CKD) status.

Results: Compared with stable 1-year NT–proBNP levels, a 25% decrease in NT–proBNP was associated with a slower decrease in eGFR in participants with CKD (adjusted difference, 1.09%/y; 95% Cl, 0.35-1.83) and without CKD (adjusted difference, 51%/y; 95% Cl, 0.21-0.81; P = 0.4 for interaction). Meanwhile, a 25% increase in NT–proBNP in participants with CKD was associated with a faster decrease in eGFR (adjusted difference, -1.04%/y; 95% Cl, -1.72 to -0.36) and risk of a 30% decrease in eGFR (adjusted odds ratio, 1.44; 95% Cl, 1.06-1.96); associations were stronger in participants with CKD than in participants without CKD (P = 0.01 and P < 0.001 for interaction, respectively). Relationships were similar irrespective of the randomized BP arm in SPRINT (P > 0.2 for interactions).

Limitations: Persons with diabetes and proteinuria >1 g/d were excluded.

Conclusions: Changes in NT–proBNP during BP treatment are independently associated with subsequent kidney function decline, particularly in people with CKD. Future studies should assess whether routine NT–proBNP measurements may be useful in monitoring kidney risk during hypertension treatment.

PLAIN-LANGUAGE SUMMARY

N-terminal pro–B-type natriuretic peptide (NT–proBNP) is a biomarker in the blood that reflects mechanical stress on the heart. Measuring NT–proBNP may be helpful in assessing the risk of long-term losses of kidney function. In this study, we investigated the association of changes in NT–proBNP with subsequent kidney function among individuals with and without chronic kidney disease. We found that increases in NT–proBNP are associated with a faster rate of decline of kidney function, independent of baseline kidney measures. The associations were more pronounced in individuals with chronic kidney disease. Our results advance the notion of considering NT–proBNP as a dynamic tool for assessing kidney disease risk.

Graphical Abstract



In nondiabetic individuals with hypertension and at high risk of cardiovascular disease (CVD), the Systolic Blood Pressure Intervention Trial (SPRINT) demonstrated that targeting a systolic blood pressure (SBP) of <120 mm Hg compared with <140 mm Hg led to significant reductions in CVD events and all-cause death.¹ However, the effect of lower SBP targets on kidney outcomes was less clear. Although randomization to intensive SBP

lowering in SPRINT led to greater dereases in estimated glomerular filtration rate (eGFR) in the first months of the trial, these acute changes are most likely related to hemodynamic changes rather than intrinsic kidney injury. However, even after these acute changes, participants in the intensive SBP lowering arm experienced a greater risk of incident chronic kidney disease (CKD) and a slightly faster decrease in eGFR compared with the standard am.^{2–6} Given the CVD and mortality benefits of intensive SBP lowering and the central role hypertension is believed to play in the development and progression of CKD, novel approaches are needed to understand kidney disease risk during hypertension treatment.

Subclinical elevations in N-terminal pro–B-type natriuretic peptide (NT–proBNP), a measure of cardiac wall stress and neurohormonal activation, are common in the general population and have strong associations with CVD and early death.^{7–9} Previous studies have shown that single NT–proBNP measurements are associated with increased risk of incident CKD and CKD progression.^{10–25} However, it is unknown whether changes in NT–proBNP levels during hypertension treatment can aid in the assessment of kidney disease risk.

In this ancillary study of SPRINT, we evaluated associations of baseline and 1-year changes in NT–proBNP with subsequent kidney function decline. We also evaluated whether these associations varied by baseline CKD status and randomization to intensive versus standard SBP lowering. We hypothesized that higher baseline levels and greater 1-year increases in NT–proBNP would be associated with faster decreases in eGFR independent of clinical characteristics, randomized treatment arm, and baseline eGFR and albuminuria.

Methods

Study Design

The design and protocol of SPRINT have been reported previously.^{1,26} In brief, SPRINT was an National Institutes of Health–funded, open-label clinical trial that randomized participants with hypertension to an intensive SBP target of <120 mm Hg versus a standard SBP target of <140 mm Hg, with individual patient management at the discretion of the trial investigators. Inclusion criteria were age at least 50 years, SBP 130–180 mm Hg, and high CVD risk (defined as prior clinical or subclinical CVD other than stroke, CKD [eGFR 20– 59 mL/min/1.73 m²], age 75 years, or 10-year CVD risk >15% based on Framingham risk score). Key exclusion criteria included diabetes mellitus, prior stroke or transient ischemic attack, eGFR <20 mL/min/1.73 m², symptomatic heart failure, or left ventricular ejection fraction <35%. A total of 9,361 participants were enrolled between November 2010 and March 2013 across 102 sites in the United States and Puerto Rico. The SPRINT protocol included a baseline visit and follow-up visits monthly for the first 3 months and then every 3 months thereafter. The trial was stopped early on the recommendation of the data and safety monitoring board, which noted substantive evidence of treatment benefits for CVD events and mortality during their regularly scheduled interim evaluation of the data.

Baseline and 12-month concentrations of NT–proBNP were measured in 8,027 SPRINT participants. We excluded 22 participants without at least one follow-up eGFR measurement after the baseline measurement. SPRINT was approved by the institutional review boards at each participating study site, and all participants provided written informed consent.

This ancillary study was approved by the institutional review boards at the University of Texas Southwestern Medical Center; the University of California, San Francisco; the San Francisco Veterans Affairs Health Care System; and the Veterans Affairs San Diego Healthcare System.

Exposure of Interest

Blood specimens were collected at the baseline and 12-month study visits in serum separator tubes, processed immediately, and stored at -80°C until NT-proBNP measurement was performed at the SPRINT Central Laboratory (University of Minnesota, Minneapolis, MN). NT-proBNP was measured from freshly thawed serum samples using an electrochemiluminescence immunoassay on the Cobas 6000 platform (Roche Diagnostics) as previously described.²⁷ The NT-proBNP assay has interassay coefficients of variation of 2.9% at 140.3 pg/mL and 2.7% at 4,563 pg/mL, with a lower limit of detection of 5 pg/mL. Three percent of NT-proBNP levels were below the lower limit of detection; we assigned these measurements a value of 3.5 pg/mL, equivalent to the lower limit of detection divided by the square root of 2. Consistent with our previous work, baseline NT-proBNP was modeled as a continuous log-linear predictor and according to sex-specific tertiles, and the 1-year change in NT-proBNP was categorized as a 25% decrease, 25% increase, or <25% change (ie, stable) relative to the baseline NT-proBNP level on the original scale (ie, not log-transformed).^{27,28} For analyses using 1-year change in NT–proBNP as the exposure, the 12-month study visit was used as the starting point for annualized eGFR slope and 30% decrease in eGFR (Fig SI).

Outcomes

The primary outcome of interest was annualized percentage change in eGFR, which was estimated from a linear mixed-effect model based on serial serum creatinine measurements collected at each monthly visit for the first 3 months and then every 3 months thereafter. Participants were followed until death or the last available study visit before the trial was stopped in August 2015. The secondary outcome was a 30% decrease in eGFR. Serum creatinine was measured at the SPRINT Central Laboratory using an enzymatic creatinine method traceable to isotope dilute mass spectrometry (Roche). Estimated GFR was calculated by the 2021 CKD-EPI (Chronic Kidney Disease Epidemiology Collaboration) equation for creatinine.²⁹

Covariates

Age, sex, race, ethnicity, medical history, medications, and smoking status (current, former, or never) were obtained by questionnaire. Trained study coordinators measured blood pressure (BP) with an automated oscillometric device (Model 907; Omron Healthcare) according to a standardized protocol and recorded BP as the mean of 3 seated BP measurements taken 1 minute apart after a 5-minute rest period.³⁰ Body mass index was calculated as weight in kilograms divided by height in meters squared. Fasting serum total cholesterol, high-density lipoprotein cholesterol, triglycerides, and urine albumin and creatinine were measured at the SPRINT Central Laboratory.

Statistical Analyses

Descriptive statistics for baseline characteristics are reported as mean (standard deviation), median (interquartile range), or number (percentage) by sex-specific tertiles of baseline NT– proBNP. NT–proBNP was log₂-transformed to correct its right-skewed distribution.

Linear mixed-effect models with random intercepts, random slopes, and an exchangeable covariance structure were used to evaluate the associations of baseline NT-proBNP and 1-year changes in NT-proBNP levels with annualized eGFR slope. Fixed effects in the models include NT-proBNP, time, and the interactions between NT-proBNP and time, whereby the parameters of time and the interactions represent the annualized eGFR slope. An exchangeable correlation structure assumes that the correlation between any two observations within the same individual is the same regardless of the specific time points or conditions at which the measurements were taken. This assumption simplifies the correlation structure by assuming a constant correlation within individuals, and it is commonly used when there is no prior knowledge or specific information about the correlation patterns within individuals. The linear mixed-effect models used all available eGFR measures for each subject (median number of eGFR measures, 10 [IQR, 9-11]). To allow interpretation of annualized eGFR slope as a percentage, eGFR was log-transformed. Logistic regression models were used to evaluate associations of baseline NT-proBNP and 1-year changes in NT-proBNP levels with a 30% decrease in eGFR. SPRINT participants were followed until death or the last available follow-up before the trial was stopped in August 2015.

Models constructed for each outcome were adjusted for the following baseline potential confounders: demographic characteristics (age, sex, race, and ethnicity), randomization arm, kidney disease risk factors (body mass index, smoking status, prevalent CVD, baseline SBP, baseline diastolic BP [DBP], number of antihypertensive medications, diuretic use, and angiotensin-converting enzyme inhibitor or angiotensin-receptor blocker use), baseline eGFR, and baseline urine albumin-creatinine ratio. Analyses of 1-year change in NT– proBNP as the exposure adjusted for the baseline NT–proBNP level. In addition, we adjusted for the first-year change in eGFR and the first-year change in SBP to determine whether the associations of change in NT–proBNP with subsequent change in eGFR were independent of concurrent changes in eGFR and SBP. We also evaluated for interactions by baseline CKD status (eGFR <60 mL/min/1.73 m²), albuminuria (urine albumin-creatinine ratio 30 mg/g), age, sex, prevalent CVD, baseline NT–proBNP level, and randomization arm in multivariable adjusted models using likelihood ratio tests. P values for interactions were adjusted for multiple testing by using the Benjamini-Hochberg procedure and setting the false discovery rate to 5%.³¹

All analyses were conducted using Stata Statistical Software (release 13; StataCorp LP) and SPSS Statistics for Windows (version 26.0; IBM Corp).

Results

Among the 8,005 SPRINT participants included in this analysis, mean age was 68 ± 9 years, 36% were women, and mean baseline eGFR was 74 ± 19 mL/min/1.73 m². The median NT–proBNP levels were 86 pg/mL at baseline and 82 pg/mL at year 1. Relative to

baseline NT–proBNP levels, year-1 NT–proBNP levels remained stable in 2,362 participants (30%), increased by 25% in 2,804 (35%), and decreased by 25% in 2,839 (35%). SPRINT participants with higher NT–proBNP levels at baseline were older, more likely to be White, and had higher SBP and lower DBP, more prevalent CVD, lower eGFR, and greater albuminuria (Table SI). Compared with participants with stable 1-year NT–proBNP levels, those with increases in NT–proBNP had similar baseline characteristics, and those with decreases in NT–proBNP were younger, less likely to report White race, and had higher SBP, DBP, and eGFR (Table 1). Baseline CKD was present in 1,958 (24%) participants. Significant interactions by CKD status were identified for associations between baseline NT–proBNP and change in eGFR (P = 0.003 for interaction) and between increases in NT–proBNP and subsequent change in eGFR (P = 0.01 for interaction). Thus, all further analyses are reported stratified by CKD status.

Among 1,958 participants with baseline CKD and 6,047 without CKD, mean baseline eGFRs were 47 ± 10 and 82 ± 13 mL/min/1.73 m², respectively, and annualized changes in eGFR during a median of 3.3 years of follow-up were -1.37%/y (95% Cl, -1.54 to -1.20) and -1.24%/y (95% Cl, -1.31 to -1.17), respectively. Median NT–proBNP levels at baseline among participants with and without CKD were 170 pg/mL and 65 pg/mL, respectively. A 25% increase or 25% decrease in NT–proBNP corresponded to larger absolute NT–proBNP changes in the CKD subgroup (Table S2).

In the CKD and non-CKD groups, the rate of decrease in eGFR was incrementally faster from the lowest to the highest tertile of baseline NT–proBNP (Fig 1). After multivariable adjustment, higher baseline NT–proBNP levels were independently associated with a faster decrease in eGFR, with a stronger association among participants with CKD at baseline (adjusted difference in annualized eGFR slope per 2-fold higher baseline NT–proBNP level: -0.44%/y; 95% Cl, -0.56 to -0.33) than in those without CKD (adjusted difference: -0.10%/y; 95% Cl, -0.14 to -0.05; P = 0.003 for interaction). A similar pattern of results was observed using tertiles of baseline NT–proBNP (Table 2).

We next modeled associations between 1-year NT–proBNP changes and subsequent change in eGFR. Participants with CKD with stable, 25% increased, and 25% decreased NT– proBNP all had small changes in eGFR from baseline to year 1 (Table S3). However, after year 1, annualized eGFR slope was slowest in the group with a 25% decrease in NT– proBNP and fastest among those with a 25% increase (Fig 2). Compared with stable NT– proBNP levels, a 25% decrease in NT–proBNP was associated with a significantly slower decrease in eGFR in the CKD and non-CKD groups after year 1 (Table 3). Meanwhile, a 25% increase in NT–proBNP was associated with a significantly faster decrease in eGFR

in the CKD group, but this was not evident in the non-CKD group (P = 0.01 for interaction).

There were 1,552 (19.4%) participants who experienced a 30% decrease in eGFR. Among those who did not experience a 30% decrease in eGFR, only 155 (2%) died during follow-up. Higher baseline NT–proBNP levels were independently associated with a greater risk of a 30% decrease in eGFR, with stronger associations observed in those with baseline CKD (Table S4). A 25% decrease in NT–proBNP appeared to be associated with lower risk of a 30% decrease in eGFR after year 1 in the non-CKD group (odds ratio, 0.85; 95% Cl,

0.72–1.01), but not in the CKD group (odds ratio, 0.95; 95% Cl, 0.67–1.33). Conversely, a 25% increase in NT–proBNP was associated with greater risk of a 30% decrease in eGFR in the CKD group (odds ratio, 1.44; 95% Cl, 1.06–1.96), but not in the non-CKD group (odds ratio, 1.02; 95% Cl, 0.87–1.20; P < 0.001 for interaction).

Consistent with the observed interactions by CKD status, higher baseline NT-proBNP levels were more strongly associated with a faster decrease in eGFR among participants with a urine albumin-creatinine ratio 30 mg/g versus < 30 mg/g (P = 0.003 for interaction; Table S5). The association of 1-year decreases in NT-proBNP (vs stable NT-proBNP) with subsequent decreases in eGFR also appeared stronger among participants with a urine albumin-creatinine ratio 30 mg/g, although die interaction was not statistically significant (P = 0.1 for interaction; Table S5). Across the intensive and standard SBP lowering groups, annualized eGFR slope after year 1 was similarly slowest in those with a 25% decrease in NT-proBNP (Fig S2). None of the NT-proBNP associations with annualized eGFR slope and risk of a 30% decrease in eGFR were modified by randomized treatment assignment (P 0.1 for all interactions). Participants with a baseline NT-proBNP level 125 pg/mL had faster annualized decreases in eGFR after year 1 across all 1 - year NT-proBNP change categories compared with those with a baseline NT-proBNP level <125 pg/mL (Fig S3). However, the associations of a 25% increase or a 25% decrease in NT-proBNP with a subsequent decrease in eGFR were similar irrespective of baseline NT-proBNP level (P 0.1 for all interactions). NT-proBNP associations with eGFR decrease also did not vary by age, sex, or prevalent CVD (P 0.1 for all interactions).

Discussion

In this analysis of SPRINT that included repeated NT–proBNP measurements in more than 8,000 participants, higher baseline NT–proBNP levels and greater 1-year increases in NT– proBNP levels were associated with subsequent decreases in eGFR independent of clinical characteristics, randomized treatment assignment, and baseline eGFR and albuminuria. These associations were stronger among participants with CKD than in those without CKD.

Previous studies have shown that higher single NT–proBNP measurements are associated with more rapid kidney function decline and with incident CKD among individuals without CKD^{11,13} and with the risk of CKD progression and kidney failure among those with CKD.^{16–19,21,22} The present analysis expands these findings by demonstrating that: 1) longitudinal changes in NT–proBNP levels are also associated with subsequent changes in eGFR; 2) these findings appear particularly strong in those with prevalent CKD; and 3) baseline NT–proBNP and changes in NT–proBNP levels have prognostic value for kidney function decline regardless of the intensity of SBP lowering. The stronger associations in the CKD subgroup may be explained in part by higher baseline NT–proBNP levels and correspondingly greater absolute NT–proBNP changes. The NT–proBNP associations and interactions by CKD status were robust to adjustment for baseline and 1-year changes in eGFR, and similar interactions were observed in persons with and without albuminuria. The strength and consistency of these findings makes it unlikely that these findings are explained by chance, regression to the mean, or confounding by reduced NT–proBNP clearance in the setting of CKD.

Subclinical elevations in NT–proBNP may reflect chronic neurohormonal activation and venous congestion, which are mechanisms that can also contribute to the progression of kidney disease by impairing intrarenal blood flow.^{32–34} However, the complex, bidirectional interplay between the heart and kidney and the reduced NT–proBNP excretion in the setting of CKD make it difficult to attribute an individual's elevated NT–proBNP level to a specific pathophysiological process. We recently demonstrated in SPRINT that intensive SBP lowering leads to greater reductions in NT–proBNP, and that this is primarily mediated by reductions in SBP.²⁸ Here we show that 1-year increases in NT–proBNP levels are associated with subsequent decreases in eGFR independent of treatment assignment and 1-year changes in SBP. Furthermore, we observed that decreases in NT–proBNP are associated with slower decreases in eGFR despite intensive SBP lowering having been shown to lead to decreased NT–proBNP levels and slightly faster decreases in eGFR do not fully explain our findings.

The results of this study suggest that longitudinal monitoring of NT–proBNP levels during hypertension treatment, particularly in those with CKD, may be useful for identifying individuals at higher risk for subsequent loss of kidney function. This builds upon previous work in SPRINT that showed individuals with higher baseline NT–proBNP levels derive greater benefit from intensive SBP lowering and that baseline NT–proBNP levels and changes in NT–proBNP over time provide prognostic information about the risks of heart failure and death during hypertension treatment.^{27,28} Taken together, these data suggest a plausible role for routine NT–proBNP measurements during hypertension treatment to provide ongoing risk assessment of cardiorenal outcomes.

As an ancillary study of SPRINT, the present analysis benefited from the inclusion of a large cohort of individuals with and without CKD, repeated NT–proBNP measurements, and frequent and protocol-driven eGFR and BP assessments during follow-up. These data allowed us to evaluate the degree to which associations of dynamic changes in NT–proBNP levels with subsequent declines in eGFR were independent of concurrent changes in eGFR and SBP. There are also several limitations. Because of the SPRINT design, our findings may not generalize to persons with heart failure, diabetes mellitus, eGFR <20 mL/min/1.73 m², or severe proteinuria. In addition, the relatively short follow-up period of the trial precluded the evaluation of long-term changes in kidney function.

In summary, among individuals with hypertension without diabetes, higher baseline NT– proBNP levels and greater 1-year increases in NT–proBNP were associated with subsequent decreases in eGFR independent of baseline eGFR and albuminuria and regardless of the intensity of SBP lowering. These associations are particularly pronounced in persons with CKD. In the context of prior literature, these results advance the notion of considering NT– proBNP as a dynamic tool for assessing kidney disease risk dining hypertension treatment.

Supplementary Material

Refer to Web version on PubMed Central for supplementary material.

Support:

This ancillary study was supported by National Heart, Lung, and Blood Institute grant 1R01HL144112-01 to Dr Berry; National Institute of Diabetes and Digestive and Kidney Diseases grants 5R01DK098234-08 to Drs lx and Shlipak and K24DK110427 to Dr lx, National Center for Advancing Translational Sciences/National Institutes of Health grant UL1 TR001860 and linked award KL2 TR001859 (Dr Ascher), and American Heart Association grants 14EIA18560026 to Dr lx and 936281 to Dr Ascher. Analytical reagents for NT-proBNP measurements were donated by Roche. The authors thank the participants and staff members of SPRINT, which was sponsored by the National Institutes of Health, including the National Heart, Lung, and Blood Institute; the National Institute of Diabetes and Digestive and Kidney Diseases; the National Institute on Aging; and the National Institute of Neurological Disorders and Stroke under contract numbers HHSN268200900040C, HHSN268200900046C, HHSN268200900047C, HHSN268200900048C, HHSN268200900049C, and Inter-Agency Agreement Number A-HL-13-002-001. SPRINT was also supported in part by resources and use of facilities through the US Department of Veterans Affairs. The SPRINT investigators acknowledge the contribution of study medications (azilsartan and azilsartan combined with chlorthalidone) by Takeda Pharmaceuticals International Inc. For a full list of contributors to SPRINT, please see the supplementary acknowledgment: https://www.sprinttrial.org/public/dspScience.cfm. The authors also acknowledge the support from the following Clinical and Translational Science Awards funded by the National Center for Advancing Translational Sciences: Case Western Reserve University, UL1TR000439; The Ohio State University, UL1RR025755; University of Pennsylvania, UL1RR024134 and UL1TR000003; Boston University, UL1RR025771; Stanford University, UL1TR000093; Tufts University, UL1RR025752, UL1TR000073, and UL1TR001064; University of Illinois, UL1TR000050; University of Pittsburgh, UL1TR000005; University of Texas Southwestern Medical Center, 9U54TR000017-06; University of Utah, UL1TR000105-05; Vanderbilt University, UL1TR000445; George Washington University, UL1TR000075; University of California, Davis, UL1TR000002; University of Florida, UL1TR000064; University of Michigan, UL1TR000433; Wake Forest University, UL1TR001420; and a COBRE Award funded the National Institue of General Medical Sciences to Tulane University: P30GM103337.

Data Sharing:

The data that support the findings of this study are available from the National Heart, Lung, and Blood Institute Biologic Specimen and Data Repositories and the corresponding author upon request.

References

- 1. SPRINT Research Group; Wright JT, Williamson JD, Whelton PK, et al. A randomized trial of intensive versus standard blood-pressure control. N Engl J Med. 2015;373(22):2103–2116. doi:10.1056/NEJMoa1511939 [PubMed: 26551272]
- Beddhu S, Shen J, Cheung AK, et al. Implications of early decline in eGFR due to intensive BP control for cardiovascular outcomes in SPRINT. J Am Soc Nephrol. 2019;30(8):1523–1533. doi:10.1681/ASN.2018121261 [PubMed: 31324734]
- Malhotra R, Craven T, Ambrosius WT, et al. Effects of intensive blood pressure lowering on kidney tubule injury in CKD: a longitudinal subgroup analysis in SPRINT. Am J Kidney Dis. 2019;73(1):21–30. doi:10.1053/j.ajkd.2018.07.015 [PubMed: 30291012]
- Zhang WR, Craven TE, Malhotra R, et al. Kidney damage biomarkers and incident chronic kidney disease during blood pressure reduction: a case-control study. Ann Intern Med. 2018;169(9):610– 618. doi:10.7326/M18-1037 [PubMed: 30357395]
- Beddhu S, Rocco MV, Toto R, et al. Effects of intensive systolic blood pressure control on kidney and cardiovascular outcomes in persons without kidney disease. Ann Intern Med. 2017;167(6):375– 383. doi:10.7326/M16-2966 [PubMed: 28869987]
- Cheung AK, Rahman M, Reboussin DM, et al. Effects of intensive BP control in CKD. J Am Soc Nephrol. 2017;28(9):2812–2823. doi:10.1681/ASN.2017020148 [PubMed: 28642330]
- Wang TJ, Larson MG, Levy D, et al. Plasma natriuretic peptide levels and the risk of cardiovascular events and death. N Engl J Med. 2004;350(7):655–663. doi:10.1056/NEJMoa031994 [PubMed: 14960742]
- Di Angelantonio E, Chowdhury R, Sarwar N, et al. B-type natriuretic peptides and cardiovascular risk: systematic review and meta-analysis of 40 prospective studies. Circulation. 2009;120(22):2177–2187. doi:10.1161/CIRCULATIONAHA.109.884866 [PubMed: 19917883]

- Natriuretic Peptides Studies Collaboration; Willeit P, Kaptoge S, Butterworth AS, et al. Natriuretic peptides and integrated risk assessment for cardiovascular disease: an individualparticipant-data meta-analysis. Lancet Diabetes Endocrinol. 2016;4(10):840–849. doi:10.1016/ S2213-8587(16)30196-6 [PubMed: 27599814]
- Ho JE, Hwang SJ, Wollert KC, et al. Biomarkers of cardiovascular stress and incident chronic kidney disease. Clin Chem. 2013;59(11):1613–1620. doi:10.1373/clinchem.2013.205716 [PubMed: 23873716]
- Park M, Vittinghoff E, Shlipak MG, Mishra R, Whooley M, Bansal N. Associations of N-terminal pro-B-type natriuretic peptide with kidney function decline in persons without clinical heart failure in the Heart and Soul Study. Am Heart J. 2014;168(6):931–939.e2. doi:10.1016/j.ahj.2014.09.008 [PubMed: 25458658]
- Ogawa N, Komura H, Kuwasako K, Kitamura K, Kato J. Plasma levels of natriuretic peptides and development of chronic kidney disease. BMC Nephrol. 2015;16:171. doi:10.1186/ s12882-015-0163-9 [PubMed: 26499263]
- Bansal N, Katz R, Dalrymple L, et al. NT-proBNP and troponin T and risk of rapid kidney function decline and incident CKD in elderly adults. Clin J Am Soc Nephrol. 2015;10(2):205–214. doi:10.2215/CJN.04910514 [PubMed: 25605700]
- Jenks SJ, Conway BR, McLachlan S, et al. Cardiovascular disease biomarkers are associated with declining renal function in type 2 diabetes. Diabetologia. 2017;60(8):1400–1408. doi:10.1007/ S00125-017-4297-0 [PubMed: 28528401]
- Sasaki T, Oishi E, Nagata T, et al. N-terminal pro-B-type natriuretic peptide and incident CKD. Kidney Int Rep. 2021;6(4):976–985. doi:10.1016/j.ekir.2021.01.006 [PubMed: 33912747]
- Spanaus KS, Kronenberg F, Ritz E, et al. B-type natriuretic peptide concentrations predict the progression of nondiabetic chronic kidney disease: the Mild-to-Moderate Kidney Disease Study. Clin Chem. 2007;53(7):1264–1272. doi:10.1373/clinchem.2006.083170 [PubMed: 17478561]
- Desai AS, Toto R, Jarolim P, et al. Association between cardiac biomarkers and the development of ESRD in patients with type 2 diabetes mellitus, anemia, and CKD. Am J Kidney Dis. 2011;58(5):717–728. doi:10.1053/j.ajkd.2011.05.020 [PubMed: 21820220]
- Yasuda K, Kimura T, Sasaki K, et al. Plasma B-type natriuretic peptide level predicts kidney prognosis in patients with predialysis chronic kidney disease. Nephrol Dial Transplant. 2012;27(10):3885–3891. doi:10.1093/ndt/gfs365 [PubMed: 23114906]
- Kim Y, Matsushita K, Sang Y, et al. Association of high-sensitivity cardiac troponin T and natriuretic peptide with incident ESRD: the Atherosclerosis Risk in Communities (ARIC) study. Am J Kidney Dis. 2015;65(4):550–558. doi:10.1053/j.ajkd.2014.08.021 [PubMed: 25446023]
- Yoshitomi R, Nakayama M, Sakoh T, et al. Plasma B-type natriuretic peptide concentration is independently associated with kidney function decline in Japanese patients with chronic kidney disease. J Hypertens. 2016;34(4):753–761. doi:10.1097/HJH.00000000000847 [PubMed: 26828787]
- Bansal N, Zelnick L, Shlipak MG, et al. Cardiac and stress biomarkers and chronic kidney disease progression: the CRIC Study. Clin Chem. 2019;65(11):1448–1457. doi:10.1373/ clinchem.2019.305797 [PubMed: 31578216]
- Anderson AH, Xie D, Wang X, et al. Novel risk factors for progression of diabetic and nondiabetic CKD: findings from the Chronic Renal Insufficiency Cohort (CRIC) Study. Am J Kidney Dis. 2021;77(1):56–73.e1. doi:10.1053/j.ajkd.2020.07.011 [PubMed: 32866540]
- Sundqvist S, Larson T, Cauliez B, et al. Clinical value of natriuretic peptides in predicting time to dialysis in stage 4 and 5 chronic kidney disease patients. PLoS ONE. 2016;11(8): e0159914. doi:10.1371/journal.pone.0159914 [PubMed: 27548064]
- 24. Sawada R, Hashimoto Y, Senmaru T, et al. Serum N-terminal pro-brain natriuretic peptide level is associated with the development of chronic kidney diseases in patients with type 2 diabetes. Endocr Metab Immune Disord Drug Targets. 2018;18(6):590–595. doi:10.2174/1871530318666180426112050 [PubMed: 29697035]
- Nakayama S, Satoh M, Metoki H, et al. N-terminal pro-B-type natriuretic peptide is a predictor of chronic kidney disease in an Asian general population — the Ohasama Study. Circ Rep. 2019;2(1):24–32. doi:10.1253/circrep.CR-19-0044 [PubMed: 33693171]

- 26. Ambrosius WT, Sink KM, Foy CG, et al. The design and rationale of a multicenter clinical trial comparing two strategies for control of systolic blood pressure: the Systolic Blood Pressure Intervention Trial (SPRINT). Clin Trials. 2014; 11 (5): 532–546. doi:10.1177/1740774514537404 [PubMed: 24902920]
- 27. Berry JD, Nambi V, Ambrosius WT, et al. Associations of high-sensitivity troponin and natriuretic peptide levels with outcomes after intensive blood pressure lowering: findings from the SPRINT randomized clinical trial. JAMA Cardiol. 2021;6(1 2):1397–1405. doi:10.1001/ jamacardio.2021.3187 [PubMed: 34468696]
- Berry JD, Chen H, Nambi V, et al. Effect of intensive blood pressure control on troponin and natriuretic peptide levels: findings from SPRINT. Circulation. 2023;147(4):310–323. doi:10.1161/ CIRCULATIONAHA.122.059960 [PubMed: 36533535]
- Inker LA, Eneanya ND, Coresh J, et al. New creatinine- and cystatin C-based equations to estimate GFR without race. N Engl J Med. 2021;385(19):1737–1749. doi:10.1056/NEJMoa2102953 [PubMed: 34554658]
- Johnson Karen C, Whelton Paul K, Cushman William C, et al. Blood pressure measurement in SPRINT (Systolic Blood Pressure Intervention Trial). Hypertension. 2018;71 (5):848–857. doi:10.1161/HYPERTENSIONAHA.117.10479 [PubMed: 29531173]
- 31. Benjamini Y, Hochberg Y. Controlling the false discovery rate: a practical and powerful approach to multiple testing. J R Stat Soc Ser B Methodol. 1995;57(1):289–300. doi:10.1111/ j.2517-6161.1995.tb02031.x
- 32. Damman K, Navis G, Smilde TDJ, et al. Decreased cardiac output, venous congestion and the association with renal impairment in patients with cardiac dysfunction. Eur J Heart Fail. 2007;9(9):872–878. doi:10.1016/j.ejheart.2007.05.010 [PubMed: 17586090]
- Mullens W, Abrahams Z, Francis GS, et al. Importance of venous congestion for worsening of renal function in advanced decompensated heart failure. J Am Coll Cardiol. 2009;53(7):589–596. doi:10.1016/j.jacc.2008.05.068 [PubMed: 19215833]
- 34. Rangaswami J, Bhalla V, Blair JEA, et al. Cardiorenal syndrome: classification, pathophysiology, diagnosis, and treatment strategies: a scientific statement from the American Heart Association. Circulation. 2019;139(16):e840–e878. doi:10.1161/CIR.00000000000664 [PubMed: 30852913]



Figure 1.

Baseline tertiles of N-terminal pro–B-type natriuretic peptide (NT–proBNP) and subsequent kidney function decline. Bars represent unadjusted estimated annual change in estimated glomerular filtration rate (eGFR) from baseline with 95% CIs. Results were stratified by baseline chronic kidney disease (CKD) status. Estimates are derived from linear mixed-effect models.



Figure 2.

Change in N-terminal pro–B-type natriuretic peptide (NT–proBNP) and subsequent kidney function decline. Bars represent unadjusted estimated annual change in estimated glomerular filtration rate (eGFR) from year 1 with 95% CIs. Results were stratified by baseline chronic kidney disease (CKD) status. Estimates are derived from linear mixed-effect models.

Table 1.

Baseline Characteristics of SPRINT Participants Stratified by 1-Year Changes in NT-proBNP

Characteristic	25% Increase in NT-proBNP (n = 2,804)	Stable NT-proBNP (n = 2,362)	25% Decrease in NT-proBNP (n = 2,839)	Total (N = 8,005)
Age, y	68±9	70 ± 9	66±9	68 ± 9
Female sex	986 (35%)	876 (37%)	1,039 (37%)	2,901 (36%)
Race				
White	1,692 (60%)	1,465 (62%)	1,553 (55%)	4,710 (59%)
Black	774 (28%)	605 (26%)	915 (32%)	2,294 (29%)
Hispanic	296 (11%)	250 (11%)	311 (11%)	857 (11%)
Smoking				
Current	1,230 (44%)	1,083(46%)	1,268 (45%)	4,710 (59%)
Former	1,218 (44%)	1,046(44%)	1,148 (41%)	2,294 (29%)
Never	353 (13%)	232 (10%)	420 (15%)	857 (11%)
Prevalent CVD	544 (19%)	493 (21%)	553 (20%)	1,590 (20%)
Prevalent heart failure	96 (3%)	78 (3%)	92 (2%)	266 (3%)
Systolic BP, mm Hg	137 ± 15	139 ± 15	143 ± 16	140 ± 16
Diastolic BP, mm Hg	77 ± 11	77 ± 11	80 ± 12	78 ± 12
No. of antihypertensive medications	1.9 ± 1.0	1.8 ± 1.1	1.8 ± 1.1	1.8 ± 1.0
Antihypertensive medication				
β-Blocker	925 (33%)	882 (37%)	1,120 (40%)	2,927 (37%)
Diuretic	1,373 (49%)	1,088~(46%)	1,244 (44%)	3,705 (46%)
Calcium channel blocker	1,054 (38%)	883 (37%)	871 (31%)	2,808 (35%)
ACEI or ARB	1,614 (58%)	1,290(55%)	1,563 (55%)	4,467 (56%)
BMI, kg/m ²	29.9 ± 5.7	29.5 ± 5.5	30.1 ± 5.8	29.8 ± 5.7
eGFR, mL/min/1.73 m ²	73 ± 20	72 ± 19	75 ± 19	74 ± 19
eGFR <60 mL/min/1.73 m ²	757 (27%)	592 (25%)	609 (22%)	1,958 (24%)
Urine ACR, mg/g	9 [6–20]	10 [6-22]	10 [6–21]	10 [6-21]
Data displayed as mean ± standard devi	iation, n (%), or median [IQR] as appropriate.			

Author Manuscript

Author Manuscript

Abbreviations: ACEI, angiotensin-converting enzyme inhibitor; ACR, albumin-to-creatinine ratio; ARB, angiotensin receptor blocker; BMI, body mass index; BP, blood pressure; CKD, chronic kidney disease; CVD, cardiovascular disease; eGFR, estimated glomerular filtration rate (based on creatinine); NT-proBNP, N-terminal pro-brain natriuretic peptide; SPRINT, Systolic Blood Pressure Intervention Trial. Author Manuscript

Table 2,

Associations of Baseline NT-proBNP With Annualized eGFR Slope Stratified by Baseline CKD Status in SPRINT

		Annualized eGFR Slop	De, %/y (95% Cl)
Baseline NT-proBNP	Z	Mean	Adjusted Difference
CKD			
Per 2-fold higher ^a	1,958	-1.37 (-1.54 to -1.20)	-0.44 (-0.56 to -0.33)
Tertile 1	371	0.10 (-0.27 to 0.48)	Reference
Tertile 2	571	-0.61 (-0.92 to -0.31)	-0.24 (-0.74 to 0.25)
Tertile 3	1,016	-2.45 (-2.68 to-2.21)	-1.38 (-1.90 to -0.87)
No CKD			
Per 2-fold higher	6,047	-1.24 (-1.31 to -1.17)	-0.10 (-0.14 to -0.05)
Tertile 1	2,357	-0.89 (-1.00 to -0.78)	Reference
Tertile 2	2,102	-1.24 (-1.36 to -1.12)	-0.10 (-0.27 to 0.07)
Tertile 3	1,588	-1.81 (-1.95 to-1.67)	-0.39 (-0.60 to -0.18)
Models adjust for and sev	race/eth	minity randomization arm	hacalina cardiovacoular o

cular disease, current smoking, body mass index, systolic blood pressure, diastolic blood pressure, number of antihypertensive medications, diuretic agent use, angiotensin-converting enzyme inhibitor or angiotensin receptor blocker use, baseline eGFR, and urine albumin-creatinine ratio. cumuruy, SCA. LACC us adjust for age, M

Abbreviations: CKD, chronic kidney disease; eGFR, estimated glomerular filtration rate; NT-proBNP, N-terminal pro-B-type natriuretic peptide; SPRINT, Systolic Blood Pressure Intervention Trial.

 ^{a}P = 0.003 for interaction by CKD status.

Table 3.

Associations of 1-Year Changes in NT-proBNP With Annualized eGFR Slope After Year 1 Stratified by Baseline CKD Status in SPRINT

		Annualized eGFR Slope After Year 1, %/y (95% Cl)	
1-Year NT-proBNP	Ν	Mean	Adjusted Difference
CKD			
25% decrease ^a	609	-0.25 (-0.74 to 0.24)	1.09 (0.35 to 1.83)
Stable	592	-1.05 (-1.56 to -0.55)	Reference
25% increase ^b	757	-2.05 (-2.48 to -1.62)	-1.04 (-1.72 to -0.36)
No CKD			
25% decrease	2,230	-0.16 (-0.36 to 0.04)	0.51 (0.21 to 0.81)
Stable	1,770	-0.69 (-0.91 to -0.47)	Reference
25% increase	2,047	-0.53 (-0.74 to -0.33)	0.12 (-0.18 to 0.42)

Models adjust for age, sex, race/ethnicity, randomization arm, baseline cardiovascular disease, current smoking, body mass index, diastolic blood pressure, number of antihypertensive medications used, diaretic agent use, angiotensin-converting enzyme inhibitor or angiotensin-receptor blocker use, urine albumin-creatinine ratio, baseline NT-proBNP, baseline and 1-year change in systolic blood pressure, and year-1 and 1-year change in eGFR.

Abbreviations: CKD, chronic kidney disease; eGFR, estimated glomerular filtration rate; NT-proBNP, N-terminal pro-B-type natriuretic peptide; SBP, systolic blood pressure; SPRINT, Systolic Blood Pressure Intervention Trial.

 ${}^{a}P = 0.4$ for interaction by CKD status comparing 25% decrease in NT–proBNP versus stable NT–proBNP.

 ^{b}P = 0.01 for interaction by CKD status comparing 25% increase in NT-proBNP versus stable NT-proBNP.